Author’s response to reviews

Title: Identification of key pathways and genes in PTEN mutation prostate cancer by bioinformatics analysis

Authors:
Jian Sun (18651111090@sina.cn)
Shugen Li (727161698@qq.com)
Fei Wang (285456241@qq.com)
Caibin Fan (fancaibin@sina.com)
Jianqing Wang (jqwang14@fudan.edu.cn)

Version: 2 Date: 13 Jul 2019

Author’s response to reviews:

Response letter
Dear editor,
We have carefully reviewed the reviewers’ remarks and here present point-to-point response in the revised manuscript. All revised parts are marked in red. Many thanks for your processing.